Author’s response to reviews

Title: Part II, Provider Perspectives: Should patients be activated to request evidence-based medicine? : a qualitative study of the VA Project to Implement Diuretics (VAPID)

Authors:

Colin D Buzza (colin-buzza@uiowa.edu)
Monica B Williams (monica.williams2@va.gov)
Mark W Vander Weg (mark.vanderweg@va.gov)
Alan J Christensen (alan-christensen@uiowa.edu)
Peter J Kaboli (peter.kaboli@va.gov)
Heather Schacht Reisinger (heather.reisinger@va.gov)

Version: 5 Date: 10 February 2010

Author’s response to reviews:

Dear Editors:

We have made all edits to the paper as requested. Unfortunately, reference #40 and 61 were duplicates. We've deleted both from the manuscript because the paper was not accepted for publication at the Journal of General Internal Medicine.

Please let us know if you have any further questions.

Thank you!

Heather